Author's response to reviews

Title: Dealing with suicidal patients - a challenging task: A qualitative study of young physicians experiences

Authors:
Tordis S. Hoifodt (tordis.hoifodt@unn.no)
Anne-Grethe Talseth (anne.g.talseth@hitos.no)

Version: 3 Date: 21 August 2006

Author's response to reviews:

Tromso, August 21. 2006

Dear Editor
BMC Medical Education

Dealing with suicidal patients - a challenging task: A qualitative study of young physicians experiences
MS: 2043432962103390

We hereby send a revised manuscript entitled: "Dealing with suicidal patients - a challenging task: A qualitative study of young physicians experiences" for consideration for publication in BMC Medical Education.

We have done the formatting changes requested:
* The country is included in the details for the second author
* All underlining is removed and to some extent replaced with bold text
* The manus has been checked as to typing errors etc and some small corrections have been made.

Yours sincerely

Tordis Sorensen Hoifodt